H i C N Households in Conflict Network The Institute of Development Studies - at the University of Sussex - Falmer - Brighton - BN1 9RE www.hicn.org

The Impact of Internal Displacement on Destination Communities: Evidence from the Colombian Conflict

Juan S. Morales*

HiCN Working Paper 209

February 2016

Abstract:

More than ten percent of the population of Colombia has been forced to migrate due to civil war. This study employs an enclave IV strategy, which exploits social distance between origin and destination locations, as well as conflict induced migration, to estimate the impact that the arrival of displaced individuals has on local residents. I compare the effects on four different subgroups of the population, partitioned by skill (low-skilled versus high-skilled) and by gender. The analysis suggests that a conflict-induced increase in population leads to a short-run negative impact on wages. Though the impact tends to dissipate over time, it persists for one group, low-skilled women. The arrival of internally displaced people also affects local access to public goods, I find a negative effect on access to piped water, and a positive effect on access to trash collection services.

Keywords: civil conflict; migration; labour markets; public goods

JEL codes: H41, J40, J61, O15, R23

^{*} Department of Economics, University of Toronto (e-mail: juan.morales@utoronto.ca).

I thank Gustavo Bobonis, Albert Berry, Ana María Ibañez, Marco Gonzalez-Navarro, Paola Salardi, Michel Serafinelli, Laura Turner, Hidenori Takahashi, Nicolas Gendron-Carrier, Tongtong Hao, Jean-William P. Laliberté and Palermo Penano, as well as participants of the University of Toronto CEPA brown bag seminar, the 2015 CEA annual conference, the 11th annual HiCN workshop, and the LACEA Labor Network: *Are labor markets in Latin America prepared for the new challenges?* conference, for useful comments and discussions. I also thank Gilles Duranton for sharing with me some of the data used. All errors are my own.

1 Introduction

The world is plagued by violence. Ongoing conflicts across the globe, in Iraq, Colombia, Nigeria, Libya, Mexico, Ukraine, Democratic Republic of Congo and Syria among others, and stories of migrants leaving these conflict zones, are often featured in the news. There are now over 50 million people around the world who have fled their homes as a result of armed conflict, the largest number since World War II (UNHCR, 2014). Over 38 million of these are internally displaced people who remain within the boundaries of their home countries (IDMC, 2015). In Colombia alone, more than six million people have been displaced by violence, over ten percent of the country's population. Although the direct impact that violence has on the lives of these victims is evident, the effects of displacement also extend to wider settings. As these forced migrants arrive in cities and towns, economic and political channels diffuse these effects among residents at the arriving locations. I study how the arrival of those displaced by violence between 1998 and 2005, a period which saw unprecedented levels of displacement in Colombia, affected destination communities by studying two important determinants of welfare: wages and access to public goods.

I build on previous work by Calderón-Mejía and Ibáñez (2015), who examine the impact of internal displacement on urban labour markets, and expand their analysis by looking at a wider range of receiving municipalities, allowing for longer-run effects, and studying a broader number of outcomes. To estimate the causal effects of interest I employ an instrumental variable strategy which uses migrant shares from the origin municipality to predict displacement to the destination. The instrument combines two factors, which are arguably independent of conditions at the destination, to predict migration flows after 1997: displacement driven by violence at the municipalities of origin, and shares of migrants from these origin municipalities who were living at the destination in 1993. The migration networks, or enclave, instrument has been commonly used in the immigration literature since Card (2001), but has not been previously used (to the best of my knowledge) to analyze internal migration in Colombia.

I frame the analysis within a spatial equilibrium model that outlines the mechanisms through which the welfare of residents at destination locations is affected with the arrival of forced migrants. Though the impact is initially concentrated in the receiving municipalities, out-migration from these tends to dissipate the effects over time. I find empirical support for the conceptual framework presented. Receiving municipalities are adversely affected in the short-run by a negative impact on wages, as in Calderón-Mejía and Ibáñez (2015). Over time, however, these effects tend to dissipate. I present evidence that labour reallocation may explain these observations, a result in line with Monras (2015)'s analysis of Mexican migration into the U.S. The effect on wages, however, persists for low-skilled women, suggesting that this group is particularly afflicted by the arrival of forced migrants. In addition, internal displacement has a positive impact on access to trash collection services, and a negative impact on access to piped water for local residents.

The paper is organized as follows. Section 2 reviews the relevant literature. Section 3 briefly outlines the background of civil conflict in Colombia. Section 4 presents the theoretical framework. The data and summary statistics are described in section 5. Section 6 outlines the empirical strategy. Section 7 presents the results and section 8 concludes.

2 Literature

The analysis I present contributes mainly to two important areas of study. First, there is a growing literature in economics on the causes and effects of civil war (Blattman and Miguel, 2010). Since 1960, twenty percent of countries have experienced civil conflicts lasting at least ten years, which highlights the importance of the issue for economic development and welfare. Second, the study contributes to a more traditional and extensive literature on immigration and labour markets.

Studies on internal displacement which examine how those displaced are affected include Kondylis (2010), who finds that displaced men in Bosnia are more likely to be unemployed, while displaced women are less likely to be in the workforce, as well as various studies on internal displacement in Colombia. Silva Arias and Sarmiento Espinel (2013) find that internally displaced people are around 6 percentage points more likely to be unemployed than other migrants, with the impact considerably mitigated by higher levels of education. Ibáñez and Vélez (2008) find that internally displaced people suffer welfare losses equivalent to 37% of their lifetime consumption, while Ibáñez and Moya (2010) find that only 25% of displaced households recover the asset losses generated by displacement. Calderón, Ibáñez and Gáfaro (2011) report that displaced women are more competitive in labour markets relative to displaced men, suggesting that the labour market effects for local residents may differ by gender.

Studies that analyze the impact that internal displacement has on host (destination) communities include Alix-Garcia and Saah (2010), who find that food prices in Tanzania increased following refugee inflows from Burundi and Rwanda in the 1990s, but that the impact is mitigated by food aid, and that while wealth effects are positive for rural inhabitants close to refugee camps, they are negative for those in urban areas. Baez (2011), studying the same conflict, finds worsening health outcomes for children in Tanzanian regions with larger number of refugees. Maystadt and Verwimp (2014) find that the presence of refugees in Tanzania led to increased household consumption, though the impact differs widely by occupation, while Maystadt and Duranton (2014) suggest that refugee induced road construction may have led to permanent welfare gains through a reduction in transportation costs. Alix-Garcia and Bartlett (2015) find that high-skilled employment is higher in a Darfurian city affected by displacement compared to a similar neighbouring city which was unaffected, as the arrival of internally displaced people generates a positive demand driven shock for these workers. Ruiz and Vargas-Silva (2013) conclude that the evidence regarding the impact of forced migrants on destination communities is mixed, and emphasize Maystadt and Verwimp (2014)'s view that there tend to be both winners and losers in the host communities.

It should be highlighted that while both the Tanzanian refugees and those displaced in Darfur were mostly localized in large camps, internally displaced people in Colombia are located all throughout the country. Studying the impact of displacement on Colombian labour markets, Bozzoli, Brück and Wald (2013) find that self-employment is lower in origin municipalities of displacement, and higher in destination municipalities, while Peña (2014) highlights the ability of labour markets to absorb forced migrants with negligible impacts on the employment outcomes of the native population. Calderón-Mejía and Ibáñez (2015) find that internal displacement has a large negative impact on wages in the informal sector of destination cities, and that it increases the probability of native workers working in the informal sector. Taken together, these studies suggest that informal sectors in Colombian labour markets seem to easily absorb the inflow of forced migrants, but that wages in destination communities are likely to be affected as a result. The short-run estimates I report support these conclusions.

The paper also contributes to the broader debate on the effects of immigration on labour markets. Several studies have found that such effects tend to be small (see for instance Card (2005, 2009); Friedberg (2001); Ottaviano and Peri (2012); Peri and Yasenov (2015)), suggesting that immigrant workers and native workers are imperfect substitutes. Others, however, argue that the effects can be large and are usually concentrated among young or poorly educated workers, who generally compete with (low-skilled) immigrants for jobs (see Borjas (2003, 2006, 2015); Smith (2012)). Internally displaced people within Colombia are however more likely to be similar to Colombian urban residents than, for instance, Cuban immigrants are to U.S. residents. I study how forced migrants may differentially affect natives depending on their relative skill levels. I present evidence for the case of Colombia consistent with Monras (2015)'s study of Mexican immigration to the U.S, which finds a large negative short-run impact on wages, and stresses how labour reallocation dissipates these impacts in the long-run.

Though not many studies have analyzed how the arrival of displaced individuals affects the provision of public goods of local residents, these unexplored impacts may have important implications for the welfare of receiving locations given the large magnitude of displacement in Colombia. Forero (2003) highlights the public policy tension that arises between a focus of guaranteeing full compensation to victims of the civil conflict, and one which solely seeks to satisfy their basic needs within a conventional social assistance framework. Bohada (2010) suggests that the arrival of internally displaced people has positive effects for the provision of public goods in receiving neighbourhoods in the Colombian city of Pasto, through increased government spending on health and educational facilities. Neighbourhoods where these migrants arrive seem to become more visible in the eyes of policymakers, and may therefore capture larger investments from the federal government or international organizations. I explore whether such investments crowd out public funding in other neighbourhoods or whether access to these services increases overall by studying impacts at the municipal level.

3 Background

It is not uncommon for countries to undergo civil wars and internal conflict throughout the process of economic development. Since 1958 an estimated 220,000 people have died in Colombia, over 80% of whom were civilians, due to one of the longest ongoing civil conflicts in the world (CNMH, 2013).

In the 1960s, the emergence of left-wing guerrilla groups ended a transition from violence between political parties to one of a subversive nature. The Revolutionary Armed Forces of Colombia (FARC) and the National Liberation Army (ELN), both founded in 1964, continue to play leading roles in the civil war today. It was not until the 1980s, however, that these guerrilla groups started an aggressive geographic expansion of their activities throughout the country. Sánchez Torres and Díaz (2005) document this expansion and highlight both the substantial increase in the number of attacks from these groups, and the widening of their geographic reach. Their operations are mainly funded through kidnappings, extortion, trade of illegal drugs, and as of more recent, illegal mining. Various attempts of peace negotiations between these guerilla groups and the Colombian government have yet to yield any lasting results. The most recent round of negotiations with FARC are currently ongoing.

In the 1980s, the increasing threat from guerillas and the inability of the government to ensure safety across the country led many landowners to sponsor local militia forces. These right-wing groups merged and founded the United Self-Defense Forces of Colombia (AUC) in 1997, a counter-insurgency paramilitary

group with the objective of combating the FARC and ELN. Funded principally by narcotraffickers and other private interests, the AUC were widely condemned of human rights abuses, including massacres, torture, and assassinations. After negotiations with the government of Álvaro Uribe Vélez, most AUC members demobilized between 2004 and 2006. Some remain active today as part of smaller militia and criminal groups.

The Colombian military is one of the largest in Latin America and has often received funding and support from the United States government as part of their 'war on drugs' campaign. After an unsuccessful attempt at negotiations with the FARC broke down in 2002, Álvaro Uribe Vélez was elected president running on a platform of aggressive military intervention against the rebel insurgencies. During his eight year mandate (having been re-elected in 2006), the army intensified its efforts of combating the guerilla groups and recovered a substantial share of the country's territory that was under FARC and ELN control.

All of these groups, including the Colombian military, have been accused of human rights violations. The multi-party nature of the conflict and its intensity has resulted in civilians being victims of cross-fire, threats, kidnappings, massacres, bombings, forced recruitment of minors, and extortion. Civilians are targeted and persecuted as a deliberate strategy of war, often accused of aiding enemy groups, whoever these are. Millions of peasants and farmers have opted for fleeing conflict zones to ensure their and their family's safety. These internally displaced people leave their hometowns and often arrive at cities or large towns.

Though the experiences of millions of victims of displacement are undoubtedly unique, many share common traits. Two specific settlement approaches are frequent when arriving at their destinations: *nuclear settlements*, in which vacant lots, often in the peripheries of cities, are 'invaded' by groups of displaced individuals to raise huts and shelters in which to live; and *dispersed settlement*, finding accommodation in the homes of family or friends who reside there (Naranjo, 2004). Internally displaced people are likely to become part of the informal economy of the municipalities where they arrive, often as construction workers, maids, or selling food items or other cheap goods on the streets. Ibáñez (2008) highlights two important facts regarding the issue: first, internal displacement is unequivocally a consequence of violence, and those displaced differ substantially from other types of internal migrants; second, though access to public services is higher for displaced people in the urban areas where they settle, their incomes and consumption patterns are severely negatively affected, and they often find themselves in vulnerable conditions worse than those of other poor urban residents.

Although public policy has varied across time and governments, since 1997¹ it has generally been shaped by the belief that the displaced population, having had their constitutional rights violated, should be compensated through public programs of 'positive discrimination' (Ibáñez and Moya, 2007). The availability of government resources has however resulted insufficient in meeting the demand for assistance from the victimized population (Pérez Ortiz, 2001). Furthermore, given these fiscal and administrative limitations, the 'positive discrimination' focus may also indirectly result in the worsening of public services for the non-victimized population.

¹Law 387 of 1997 established policies aimed at assisting victims of and preventing forced displacement.

4 Conceptual Framework

To investigate how the welfare of native residents of destination cities is affected by internal displacement, I present a spatial equilibrium framework in the spirit of Moretti (2011), generalized and extended to include violent conflict v as a determinant of well-being. The economy consists of a set of locations J, each producing a single tradable good. Each individual i of type k (which defines their skill and gender) is mobile and chooses location j to maximize their indirect expected utility U_{kij} , given by:

$$U_{kij} = V_{kj}(w_{kj}, r_j, a_j) + \varepsilon_{kij}$$

where $w_{kj}(N_{kj}, v_j)$ is the wage in location j for type k, $r_j(N_j, v_j)$ is the cost of housing (of which each worker demands one unit) and $a_j(N_j, v_j)$ is a measure of the local amenities captured by the individual. Finally, $\varepsilon_{kij}(s_{ij}, v_j)$ is an idiosyncratic component for i's individual preference for location j, where s represents social capital and positively affects idiosyncratic utility $(\frac{\partial \varepsilon_{kij}}{\partial s_{kij}} > 0)$. V is increasing in w and a $(\frac{\partial V_{kj}}{\partial w_{kj}} > 0, \frac{\partial V_{kj}}{\partial a_j} > 0)$, and decreasing in $r(\frac{\partial V_{kj}}{\partial r_j} < 0)$.

The components of individuals' welfare are themselves in turn functions of N_j , the number of residents of type *k* at location *j*, and v_j , the level of violence at location *j*. In particular, wages are negatively related to population $(\frac{\partial w_{kj}}{\partial N_{kj}} < 0)$, the labor demand curve is downward sloping), the cost of housing is increasing with population $(\frac{\partial r_j}{\partial N_j} > 0)$ and the relationship between amenities and *N* is ambiguous, due to the possibility of either congestion in amenities (as in Dinkelman and Schulhofer-Wohl (2015)) or of increased access to resources due to political responses (as suggested by Bohada (2010)).

Wages, amenities and individual's preferences for a specific location are negatively affected by violence $(\frac{\partial w_{jk}}{\partial v_j} < 0, \frac{\partial a_j}{\partial v_j} < 0, \frac{\partial \epsilon_{kij}}{\partial v_j} < 0, \forall k, i)$. Violence makes individuals and firms less productive, destroys amenities, and also affects individuals directly. The cost of housing is positively affected by violence $(\frac{\partial r_j}{\partial v_j} > 0)$, through destruction of the stock of housing.² Violence therefore unambiguously affects utility *U* negatively $(\frac{\partial U_{kij}}{\partial v_i} < 0, \forall k, i)$.

4.1 Effect of civil conflict on migration and impact at destination

Assume now three locations, R, T and C. I now consider the impact that civil conflict in location R can have on C's and T's residents through internal displacement, and I differentiate short-run and long-run effects. Assume (without loss of generality) that the social distance between R and T is smaller than that between R and C. In this general framework this assumption is defined through the idiosyncratic preferences of individuals: those having higher social capital in R, have higher social capital in T than in C. That is, if $s_{kiR} > s_{kiT}$, then $s_{kiT} > s_{kiC}$.

Suppose now that the level of violence in R, v_R , increases as a result of civil conflict. As outlined above, an increase in v_R will negatively affect the utility that individuals receive from locating at R, U_{kiR} , decreases. This reduction in the utility received from R results in out-migration from location R. Displaced individuals from R relocate to T in the short-run. The arrival of migrants in location T (an increase in N_T)

²Note however, that the overall effect on the cost of housing is likely to be negative as violence induces out-migration, therefore reducing the demand for housing.

affects the utility of *T*'s residents, U_{kiT} , through three specific mechanisms: i) a change in wages w_{kT} ; ii) a change in the availability of amenities a_T ; and a change in the cost of housing r_T .³

Furthermore, if U_{kiT} is negatively affected by the arrival of displaced individuals, then the residents of *T* may in turn out-migrate to location *C*, therefore affecting U_{kiC} through an increase in N_C . Thus we can expect the impact of displacement on *T* to be mitigated by out-migration in the longer-run (as N_T decreases). It should thus be noted that the welfare impact of internal displacement is mitigated with more than two locations, firstly because the number of forced migrants arriving at each location is reduced, and in addition native residents of a location affected by the arrival of migrants may in turn choose themselves to move to a third location.

These effects will be evaluated empirically in the following sections. I estimate the welfare impact of internal displacement on native workers by analyzing the effect that the arrival of migrants has on their utility through two of the channels outlined: wages and the availability of local amenities. In particular, I aim to find estimates of $\frac{\partial w_{kj}}{\partial N_j}$ and $\frac{\partial a_j}{\partial N_j}$.⁴ I partially estimate the impact on amenities by looking at the provision of public goods.

5 Data and Summary Statistics

The main sources of data used are: a labour survey carried out by the National Administrative Department of Statistics (DANE), a government entity in charge of the collection and diffusion of Colombia's official statistics; the Colombian censuses of 1993 and 2005, also collected by the DANE; and the Unique Registry for Displaced Population (RUPD) maintained by the non-profit organization Acción Social.⁵

The survey used is the National Household Survey (ECH) for the years 1998 to 2004, a repeated crosssection collected quarterly by the DANE throughout the country (267 municipalities). The ECH includes data on wages as well as employment and schooling status. Unfortunately, these surveys do not allow for the consistent identification of displaced households, thus I assume that displaced households do not enter the sampling frame the year after their arrival.⁶ The sample is restricted to individuals between 15 and 70. The log of real wages are winsorized at the 1% level. Table 1 summarizes statistics for individuals in the ECH survey by education (less than secondary and secondary and above).

The Colombian censuses of 1993 and 2005 were accessed through the Integrated Public Use Microdata Series (IPUMS, 2014). Although the census covers the entire population of the country, the public use version covers ten percent of the population (over four million observations) across 533 municipalities (merged to be larger than 10,000) located in 31 different departments (a political division equivalent to states or provinces). The 2005 census includes detailed questions on migration, in particular, it asks people whether they have moved in the previous five years, and what the reasons were for such move. I classify as internally displaced those who list "violence or insecurity" as their reason for moving.

Household skill groups, which are pivotal to the analysis, are defined such that high-skilled households are those in which all adult members have secondary education or above. This definition provides groups

³Although wages are affected differentially by migration from households of different types, all households share the same amenities and rental markets.

⁴The model assumes that individuals of all types share rental and amenities markets, but this assumption will be evaluated empirically in section 7.

⁵Accessed through the Integrated System for Humanitarian Information for Colombia (SIDIH).

⁶The ECH's module on migration was only carried out during the first quarter of the year and was unavailable to the public at the time of this analysis.

with a roughly balanced number of households. Table 2 summarizes statistics for native households in the census (those who have not moved in the previous five years) by skill groups, as well as by IDP status. We can observe that internally displaced people are on average younger, have larger families and are more likely to belong to an ethnic minority. Their education level is on average lower than that of the native population.⁷ As expected, high-skilled households are more likely to have children in school, and have higher access to water and electricity than low-skilled households. Unsurprisingly, displaced households in the census fare poorly by all these metrics, for the most part worse than low-skilled households. In addition, summary statistics of a wealth index created using household assets and dwelling characteristics is also reported.

Over 700,000 people were registered as displaced during the period of study. The independent variable of interest is built using data from the RUPD, a unique registry which documents the number of internally displaced people both arriving at and leaving from Colombian municipalities.⁸ Note however that these are two separate counts, and there is no matching between the origin and destination of the migrants in the data. I exploit both of these data records in my empirical strategy outlined below. Various municipality controls are also added and constructed using the 1993 census. I summarize these as well as the dependent variables in Table 3. Figure 1 shows the number of forced migrants arriving in Colombia's 13 largest cities between 1997 and 2011, as well as the total number of displaced people flowing outwardly from all municipalities. In addition, Figure 2 maps the number of internally displaced people arriving at Colombian municipalities for the period preceding the census (2000-2004).

6 Empirical Strategy

Displacement due to violence presents a natural experiment to analyze important migration related issues. The flow of internally displaced people arriving at destination cities as a share of the local population provides plausibly exogenous variation in the supply of labour and the demand for public resources. The analysis will explore the impact of internal displacement separately for low-skilled and high-skilled individuals, as well as by gender. The sample is thus partitioned and all of the regressions are separately run on each of these groups, allowing for heterogeneous impacts of the covariates, by skill and gender group, on the outcomes of interest.

6.1 Short-run effects

The short-run analysis focuses on the impact of displacement on wages, using the ECH survey. The impact of forced migration on individual wages at the destination location is estimated using the following equation:

$$y_{imt} = \alpha + \beta d_{mt-1} + \lambda_i X_{imt} + \lambda_m X_{mt} + \gamma_t + \delta_m + \delta_m T + \varepsilon_{imt}$$

where y_{imt} is the log of wages for individual *i* in municipality *m* at time *t*. The inflow of internally displaced people for each one hundred people is represented by d_{mt} , that is:

$$d_{mt} = \frac{100}{pop_{mt}} \times f_m$$

⁷Silva Arias and Guataquí Roa (2008) and Ibáñez (2008) also highlight the relatively low educational attainment of internally displaced people.

⁸Displacement is known to be underreported in the census, see Silva Arias and Sarmiento Espinel (2013) for a discussion of how these two sources compare.

where f_{mt} is the total number of forced migrants arriving at municipality *m* at time *t*. The vector X_i includes individual controls (education, potential experience and its square, gender and marital status), and X_{mt} includes the log of total population. I include municipality fixed effects, year fixed effects, and municipality specific linear time trends. Note that β constitutes the parameter of interest, which estimates the impact of the arrival of forced migrants at time t - 1 on wages at time *t*. The model is estimated in two steps, first regressing wages on the individual covariates and averaging the residuals at the municipal-year level, and then regressing these average wage residuals on the municipal and time varying regressors (including displacement).

6.2 Long-run effects

An alternate specification which captures long-run effects is also studied:

$$y_{im} = \alpha + \beta d_m + \lambda_i X_{im} + \lambda_m X_m + \delta_{D_m} + \varepsilon_{im}$$

where y_{im} is the outcome of interest, X_{im} is a vector of household and individual controls (various demographic characteristics including household size, proportion of female members, gender of household head, education of household head, number of children, number of adults, proportion of adults at different levels of education, among others), X_m is a long vector of municipality level controls (which include population, longitude, latitude, altitude, share urban population, share of population who are worker migrants - that is, they list "work" in the census as the reason for migration - as well as the mean of various outcomes in 1993 - including for most cases \bar{y}_{1993m} , and δ_{D_m} are department fixed effects (where municipality *m* is located in department D_m). These department fixed effects capture variation at the department level which could affect both the probability of internal displacement and other outcomes of interest. Capturing the homogeneity of outcomes within departments is important given the possibility of spatial correlation (as highlighted by Figure 2). Finally, the inflow of internally displaced people in the previous five years for each one hundred people is represented by d_m , that is:

$$d_m = \frac{100}{pop_m} \times f_m$$

where f_m is the total number of forced migrants arriving at municipality m in the previous five years, and pop_m is the population of the municipality. The long-run analysis is done for wages using the 2004 ECH sample, and for other outcomes of interest using the 2005 census. Since the census includes detailed migration information, the sample of analysis selected in this case includes only native residents at the destination municipalities, defined as those who lived in the same municipality 5 years prior to the census.

6.3 Enclave IV

As argued by previous work, displacement is driven by episodes of violence (See Ibáñez (2008), and Calderón-Mejía and Ibáñez (2015)) and the timing of these expulsions is thus likely independent of outcomes at receiving locations. The forced migrants' choice of location may however be affected by, or correlated with, certain city characteristics. For instance, if labour market conditions are relatively favourable at a specific city at the time of departure, it is likely to attract more migrants. More specifically, the number

 $^{^{9}}$ The average level of *y* at municipality *m* in 1993, including access to schooling, access to water supply and trash collection services, unemployment, share employed in agriculture, among others.

of internally displaced people arriving, f_{mt} , could be correlated with ε_{imt} . To overcome this concern, I employ the migration networks, or enclave, instrument commonly used in the immigration literature (see Card (2001), or more recently, Monras (2015) and Lafortune, Tessada and Lewis (2014)).

Patterns of migration are often determined by past migration waves, and internally displaced people often arrive at the homes of relatives or friends (Naranjo, 2004). I define the instrument using the 1993 census. It is defined as:

$$dIV_{mt} = \frac{100}{p_0 p_{mt}} \times (\sum_j e_{jt} \times s_{mj1993})$$

where e_{jt} is the total number of expulsions from municipality *j* at time *t*, and s_{mj1993} is the share of migrants from municipality *j* who lived in municipality *m* in 1993.¹⁰ The instrument depends on migration decisions which took place prior to 1993, and is thus potentially uncorrelated with outcomes during the period of study, conditional on a wide range of past municipal controls.¹¹

The empirical strategies presented use variation in the arrival of forced migrants across time in the ECH survey to identify the short-run effects of internal displacement, an analysis similar to that of Calderón-Mejía and Ibáñez (2015). On the other hand, the analysis exploits variation across municipalities to identify the longer term impact of internal displacement, with t defined in the instrument as the period five years prior. This second strategy, IV with a wide set of controls, is similar to that in Duranton (2014)'s analysis of agglomeration effects in Colombia.

6.4 Out-migration

The long-run analysis I have presented uses migration information from the census to identify non-movers since, as predicted by the framework presented in section 4, a negative welfare impact to the residents of certain location will result in out-migration from it (it is such a shock that leads to internal displacement in the first place). As forced migrants arrive at their destinations they affect the welfare of local residents through various outcomes, if such an impact is negative overall, it would in turn also result in migration away from these receiving locations. Such process of labour reallocation has been documented in studies of immigration into the U.S, but has not, to the best of my knowledge, been studied for the case of Colombia. I use migration information from the census to examine such a possibility through the following short-run specification:

$$y_{mt} = \alpha + \beta d_{mt-1} + \lambda y_{mt-1} + \gamma_m + \delta_m + \varepsilon_{mt}$$

where y_{mt} is the rate of out-migration defined as $y_{mt} = (n_{mt}/pop_{m2000}) * 100$, n_{mt} is the number of people who moved away from municipality m at time t by relevant group (by skill or gender), and pop_{m2000} is the estimated relevant population of m in the year 2000 (by skill or gender group); d_{mt-1} is the number of internally displaced people arriving at m at time t - 1 as previously defined; I also include municipality and time fixed effects. The out-migration rates are defined for the subpopulations of low-skilled and high-skilled households by gender of the head. I include a lagged dependent variable as a control variable as violence shocks could be correlated across space and time, such that d_{mt-1} could be potentially correlated with y_{mt-1} .

¹⁰Migrants in these case are defined as people who were born in *j* but did not live there in 1993. Note that $\sum_{m} s_{mj1993} = 1$ for each *j*.

j. ¹¹Though some displacement due to violence had already taken place by 1993, the 1985 census (Colombia's previous census) does not include information on migrant's place of birth, thus the instrument cannot be constructed using earlier dates.

Finally, I study long-run effects on out-migration through the following specification:

$$y_{m2005-2000} = \alpha + \beta d_m + \lambda y_{m2000-birth} + \lambda_m X_m + \delta_{D_m} + \varepsilon_m$$

where $y_{m2005-2000} = (n_{m2005-2000}/pop_{m2000}) * 100$, $n_{m2005-2000}$ is the number of people who moved away from municipality between 2000 and 2005, and pop_{m2000} is defined as above. In addition, I control for previous rates of out-migration using $y_{m2000-birth} = (n_{m2000-birth}/pop_{m2000}) * 100$, where $n_{m2000-birth}$ are the number of people born in *m* who lived elsewhere in 2000.¹²

As before, the enclave instrument is used to predict displacement (d_{mt} and d_m) in both the short-run and long-run out-migration specifications.

7 Results

7.1 First stage

I first present estimates from regressing the number of internally displaced people as a share of the population of the destination location on the constructed instrument. The first stage estimates for the ECH surveys are presented in Table 4, and those for the census are presented in Table 5. As observed, the enclave instrument is a good predictor of arrivals at the destination municipalities, both in the short-run and in the long-run, and for both the low-skilled and the high-skilled household samples separately. Note that running the first stage regressions separately for low-skilled and high-skilled households does not alter the sample of municipalities¹³, only the relative weights that each municipality receives depending on their skill composition. It is thus not surprising that the first stage coefficients are very similar across specifications.

7.2 Impact on labour markets

7.2.1 Short-run effect on wages

Table 6 shows the results from the short-run analysis, which includes municipal fixed effects, year fixed effects and municipal specific linear trends, with the log of real wages as the outcome of interest, winsorized at the 1 and 99 percentiles. Both the OLS and the IV estimates suggest that displacement has a negative effect on local wages. The IV estimates are larger in magnitude indicating that the OLS estimates may be biased possibly due to the selection of more favourable municipalities by those displaced, as discussed above. The IV estimate for all households (column 4) indicates that a one percent conflict induced increase in population leads to a 1.4 percent reduction in wages the following year. Though the coefficient estimate is larger in magnitude for low-skilled households, it is not precisely estimated.

Table 7 further partitions the sample by both gender and skill. The IV estimates suggest that the impact of displacement on local wages is larger for women than for men, consistent with Calderón, Ibáñez and Gáfaro (2011)'s observation that displaced women may be more competitive in urban labour markets than displaced men¹⁴. A one percent increase in population due to displacement leads to a 2.2 percent reduction

¹²This control was chosen given data limitations from the census. Data is available for municipality of residence five years prior (in 2000) and municipality of birth.

¹³With the exception of one municipality which does not contain any high-skilled households in the ECH sample.

¹⁴Note that the ECH sample consists of 267 large municipalities.

of wages for low-skilled women (column 7) and a 1.7 percent reduction in wages for high-skilled women (column 8). The effect is smallest for high-skilled men, leading to a wage reduction of 0.7 percent, not statistically significant (column 6).

The estimates presented are relatively large compared to previous work on labour markets and migration.¹⁵ The large magnitude may be partly attributed to the fact that the dependent variable is defined as a share of *total* population, whereas previous literature defines it as a share of the relevant population (ie. skill group or occupational group). Because of data limitations, a direct comparison is not feasible. Note however that the short-run estimates from this section use a similar dataset and approach to Calderón-Mejía and Ibáñez (2015) and support the conclusions of that study.

7.2.2 Long-run effect on wages (and on wealth index)

Tables 8 shows the results of the long-run analysis, which includes department fixed effects and a long vector of municipal controls, on the log of real wages. The OLS estimates suggest that displacement is associated with higher local wages at the end of the five year period, a one percent conflict induced increase in population between 1999 to 2003, is associated with wages which are roughly 0.4 percent higher in 2004 (columns 1-3). The IV estimates suggest that the effect is not statistically significant. Table 9 repeats the analysis by skill and gender groups. The OLS estimates again suggest a positive association between displacement and wages. The IV estimates suggest that the long-run effect on wages is negative and statistically significant for low-skilled women, and not different from zero for the remaining groups. The estimates indicate that a one percent increase in population between 1999 and 2003 leads to a decrease in wages of about 1.6 percent in 2004.

A large and persistent impact on wages may have a measurable effect on the accumulation of wealth over a long period of time. Though information on wages is not available from the census, the impact of internal displacement on household income at receiving communities may be indirectly estimated by looking at asset ownership (as in Alix-Garcia and Saah (2010)) or other indicators of wealth. I create an index of household wealth through principal component analysis using dwelling characteristics (number of rooms, types of walls and floors, indicators for kitchen and toilet) and asset ownership (television, radio, refrigerator, water heater, computer, washer, phone, car), and estimate the long-run impact of internal displacement on this index. The results from this analysis are presented in Tables 10 and 11. The estimates indicate that internal displacement has no statistically significant long-run effects on the wealth of local residents over the period of study. It should be noted however that, though statistically insignificant, the magnitude of the effect is largest again for low-skilled women (Table 11, column 3).

Though the initial impact on wages for most groups was negative and significant in the short-run, the effect tends to dissipate or disappear in the longer run. This result appears surprising given the relatively large short-run impact found on household wages but is congruent with previous studies of immigration into the U.S, which have stressed labour reallocation as a likely explanation for these (or the absence of) findings (see for instance Borjas (2006), or Monras (2015)). I evaluate the impact of displacement on out-migration below. However, the effect persists for low-skilled women even in the long-run analysis. In terms of wage impacts, low-skilled women are disproportionately affected by incoming displacement

¹⁵With the exception perhaps of Calderón-Mejía and Ibáñez (2015), who find that a 10 percent increase in the refugee share leads to a decrease in wages in the informal sector of 2.3 percent. Such an increase in the total refugee share, at a mean of roughly 3 percent in their sample, represents a population increase of about 0.3 percent. The comparable estimate would then suggest that a 1 leads to a decrease in wages in the informal sector of about 7.6 percent.

relative to other groups. The short-run effect is largest for them and the effect seems to persist over the five year period of analysis.

7.3 Long-run impact on public goods

I now estimate the impact of internal displacement on access to water, electricity and trash collection services using data from the 2005 census. The expected effect on local households could be negative, due to congestion of public resources, or positive, perhaps due to increased visibility and transfers from the federal government or NGOs (as in Bohada (2010)). The estimates of the analysis are presented in Tables 12-17. Tables 12 and 13 show that the impact of displacement on access to electricity is not statistically significant for any of the specifications. Tables 14 and 15 suggest that the arrival of internally displaced individuals leads to decreased access to piped water, and that the effect is particularly concentrated among high-skilled households. A one percent increase in population due to displacement leads to a 0.7 percent decrease in access to piped water for high-skilled households (Table 14, column 6). Tables 16 and 17 show that the arrival of internal displacement has a positive impact on access to trash collection services for receiving municipalities. On average, a one percent increase in population has an effect on trash collection services of between 0.6 percent for low-skilled households (Table 16, column 5) and 1 percent for high-skilled households (Table 16, column 6).

The analysis suggests that municipalities which receive larger numbers of displaced people tend to have increased access to trash collection services and reduced access to piped water relative to those which received less migrants. It should be noted that water services are usually provided by public companies, electricity in many municipalities is provided through public-private partnerships, and trash collection is sometimes outsourced by the municipal government to private companies. One possible explanation for the pattern observed is that expanding access to piped water may face more capital constraints than increasing access to trash collection, a labour intensive good which may be more easily provided given the supply shock to low-skilled labour that is the arrival of forced migrants. Municipal governments aware of these changes in relative costs may optimally shift resources away from the capital intensive public good (piped water) and towards the labour intensive public good (trash collection) in order to attain a welfare maximizing policy objective. This shift in the allocation of public goods seems to persist over the 5 year period of the analysis, despite the fact that the effect on wages dissipates for the most part. An exploration of the exact mechanisms through which these effects arise is outside of the scope of the paper.

7.4 Out-migration

Table 18 reports the results from the short-run analysis of the effect of displacement on out-migration. Following an inflow of forced migrants, we expect local residents who are negatively affected by the shock to migrate. The coefficients are all positive, suggesting that the arrival of internally displaced people has a negative impact on the welfare of local residents, therefore inducing out-migration. The IV estimates indicate that a one percent increment in population increases the rate of out-migration by about 0.2 to 0.3 people per 100 initial residents. The effect is largest for high-skilled women, and smallest for low-skilled women. The results from the long-run specification, which captures the cumulative effect of arrivals from 2000 to 2004 on the rate of out-migration during the same period, is presented in Table 19. Though the coefficients remain positive, they become statistically insignificant and smaller, except for high-skilled

women. The instrument also becomes weaker likely because the sample size decreases and there is no longer variation across time.

With the theoretical framework presented in mind, the effects from this section capture two distinct effects. As internally displaced people arrive at their destination, they affect the welfare of residents through various mechanisms, however, only those for whom the effect is large enough such that their optimal choice of location changes, will move. Thus, people who out-migrate are either those for whom the impact of displacement is large, or those who are close to marginal in their location decision to begin with. Mobility plays an important role in this. If, for instance, low-skilled women are less mobile (ie. they are not marginal in their location decision), then even a large negative effect may not be enough to induce them to move, and they are likely to suffer both short-run and long-run consequences from the arrival of new residents. Groups that are more mobile are better able to cope with the impact of displacement by migrating themselves.

7.5 Robustness checks

7.5.1 Effect on wages

The main regressions on wages are repeated in this section after restricting the sample in various ways. Table 20 restricts the sample depending on whether the municipality is one of the largest 13 Colombian cities, which were part of the Calderón-Mejía and Ibáñez (2015) study. Columns 1-3 restrict the sample to just these cities, the instrument becomes weak and the results insignificant. The enclave instrument does not seem to be a good predictor of migration to these large cities. Columns 4-6 report the results of excluding these cities from the analysis. The coefficients remain negative and close in magnitude to those previously found, suggesting that the cities are not driving the results.

The ECH surveys are not representative at the municipality level and the sample is small for some of these municipalities. Table 21 and 22 restrict the sample of municipalities depending on the size of the survey. Table 21, columns 1-3 and 4-6, excludes municipalities for which the sample size represents less than five percent and ten percent of the total population, respectively. Though the instrument becomes weaker, the effect remains negative and fairly large. Table 22 restricts the sample depending on the total number of individuals surveyed at the municipality. Columns 1-3 and 4-5, omit municipalities with less than 50 and less than 100 individuals in the sample, respectively. Again, the results remain negative and large. The results are particularly large for columns 4-6, suggesting that the main estimates may be attenuated if measurement error is large in municipalities with small samples.

Tables 23-25 restrict the years of the analysis. Across specifications the results remain negative and mostly significant. The analysis indicates that the results seem to be particularly large for the later period of study, in particular after 1999 (table 23, columns 1-3).

The short-run estimates on wages assume that internally displaced people arriving at a municipality do not enter the sampling frame the following year. I bound the estimates for the impact on wages considering the possibility of them entering the sample and two extreme case scenarios: they all enter the bottom of the wage distribution, or they all enter the top. The sample is restricted such that if the share of IDP arriving at a municipality make up x percent of the population, then the bottom or the top x percent of the wage earners are removed from that municipality. These bounded estimates are presented in Table 26. Columns 1-3 show the lower bound effect, which is negative but close to zero, and columns 4-6 show the upper

bound impact, negative, relatively large and statistically significant.

7.5.2 Long-run estimates

This section reports on robustness checks performed for the public goods analysis using the census. The main findings, an increase in the availability of trash collection services, and a decrease in access to piped water, are evaluated through various alternative specifications.

The results are presented in Tables 27 through 30. Tables 27 and 29 for low-skilled households, and tables 28 and 30 for high-skilled ones. Column 1 shows the results from the preferred specification of the analysis (presented in the sections above) for comparison. Column 2 restricts the analysis to urban households. The analysis is run on subsamples of the municipalities, excluding the 13 largest cities (from the DANE surveys, column 3), and excluding the top 5% receiving municipalities (column 4), and the top 5% municipalities with the most expulsions (columns 5). Column 6 uses an alternative measure of the migration rate defined as:

$$d_{ct} = \frac{100}{p_0 p_{c1993}} \times m_{ct}$$

given the concern that pop_{ct} is affected by internal displacement. Column 7 controls for the number of expulsions (e_{ct}) that the municipality experienced, as a proxy for levels of violence in these locations which could also affect the outcomes of interest. Column 8 uses net displacement as a share of the population (total arrivals - total expulsions) as the dependent variable.

Excluding the municipalities with the most arrivals (column 3), makes the instrument considerably weaker and the estimates more imprecise. The coefficient for water access on low-skilled households becomes positive and insignificant under this specification (Table 27). It may be the case that the effect on piped water is especially concentrated in municipalities with a large number of arrivals. However, the coefficient for high-skilled households remains large and negative (Table 28), and not significantly different from the other estimates, though imprecise.

For the most part, the results for public goods are fairly robust to these alternative specifications. Local residents experience a reduction in access to piped water and an increase in access to trash collection following an inflow of internally displaced people.

8 Conclusion

The impact that forced migration has on receiving cities is broad and multidimensional. In this paper I present a spatial equilibrium model to illustrate the mechanisms by which civil conflict induces displacement and how displacement in turn affects residents at receiving locations. I estimate the magnitude of these effects using an enclave instrument constructed with reference to previous migration studies. The instrument exploits displacement driven by violence and past waves of migrants to predict the location decision of displaced individuals.

This paper presents evidence of internal displacement having a large short-run impact on local wages across Colombian municipalities. The labour supply shock is absorbed through an initial reduction of wages, not surprising given that most displaced individuals work in the informal sector where minimum wages do not bind. More interestingly, these effects seem to dissipate in the longer-run analysis. Labour reallocation may play an important role in dissipating these negative effects, as indicated by the fact that following inflows of forced migrants, out-migration of local residents increases. This result is congruent with Monras (2015)'s recent study of Mexican immigrants in the U.S.

The negative impact on wages, however, persists in the long-run for low-skilled women. Low-skilled women may be less mobile, as suggested by the fact that they react the least in the short-run to the arrival of displaced people, in terms of their out-migration rates. This low mobility, coupled with the fact that internally displaced women adapt well to new labour markets (Calderón, Ibáñez and Gáfaro, 2011), make low-skilled local women most vulnerable to the arrival of displaced individuals, both in the short-run and in the long-run.

The arrival of internally displaced people also results in higher access to trash collection services and reduced access to piped water for local residents. Future work studying the process of public resource allocation at the municipal level could help explain the underlying incentives driving these patterns. In particular, such work could investigate how local governments choose to provide public services which vary in their labour-capital intensity following changes to relative costs, induced in this case by a labour supply shock.

Conflict-induced population displacement will have impacts on other amenities related to the wellbeing of residents at destination cities, yet not explicitly studied in the empirical analysis presented. Road congestion, pollution and crime are important factors likely to be affected by the arrival of large numbers of forced migrants, and it is not clear if the impact on these outcomes is homogenous across skill and gender groups. Finally, the impact of displacement on housing costs may be more clearly studied with alternative datasets. These issues remain important avenues for future research.

References

- Alix-Garcia, Jennifer, and Anne Bartlett. 2015. "Occupations Under Fire: The Labor Market in a Complex Emergency." Oxford Economic Papers.
- Alix-Garcia, Jennifer, and David Saah. 2010. "The Effect of Refugee Inflows on Host Communities: Evidence From Tanzania." *The World Bank Economic Review*, 24(1): 148–170.
- **Baez, Javier E.** 2011. "Civil Wars Beyond Their Borders: The Human Capital and Health Consequences of Hosting Refugees." *Journal of Development Economics*, 96(2): 391–408.
- Blattman, Christopher, and Edward Miguel. 2010. "Civil War." Journal of Economic Literature, 3-57.
- **Bohada, María del Pilar.** 2010. "Desplazamiento Forzado Y Condiciones De Vida De Las Comunidades De Destino: El Caso de Pasto, Nariño." *Revista de Economía Institucional*, 12(23): 259–298.
- **Borjas, George J.** 2003. "The Labor Demand Curve Is Downward Sloping: Reexamining the Impact of Immigration on the Labor Market." *The Quarterly Journal of Economics*, 118(4): 1335–1374.
- **Borjas, George J.** 2006. "Native Internal Migration and the Labor Market Impact of Immigration." *Journal of Human resources*, 41(2): 221–258.
- Borjas, George J. 2015. "The Wage Impact of the Marielitos: A Reappraisal." NBER Working Paper No. 21588.
- **Bozzoli, Carlos, Tilman Brück, and Nina Wald.** 2013. "Self-employment and Conflict in Colombia." *Journal* of Conflict Resolution, 57(1): 117–142.
- **Calderón-Mejía, Valentina, and Ana María Ibáñez.** 2015. "Labour Market Effects of Migration-related Supply Shocks: Evidence From Internal Refugees in Colombia." *Journal of Economic Geography*, 1–19.
- **Calderón, Valentina, Ana María Ibáñez, and Margarita Gáfaro.** 2011. "Forced Migration, Female Labour Force Participation, and Intra-Household Bargaining: Does Conflict Empower Women?" *MICROCON Research Working Paper 56.*
- **Card, David.** 2001. "Immigrant Inflows, Native Outflows, and the Local Labor Market Impacts of Higher Immigration." *Journal of Labor Economics*, 19(1): 22–64.
- Card, David. 2005. "Is The New Immigration Really So Bad?" The Economic Journal, 115(507): 300-323.
- Card, David. 2009. "Immigration and Inequality." American Economic Review, 99(2): 1-21.
- **CNMH.** 2013. *Basta ya! Colombia: Memorias de Guerra y Dignidad.* Centro Nacional de Memoria Histórica, Comisión Nacional de Reparación y Reconciliación (Colombia), Área de Memoria Histórica.
- **Dinkelman, Taryn, and Sam Schulhofer-Wohl.** 2015. "Migration, Congestion Externalities, and The Evaluation of Spatial Investments." *Journal of Development Economics*.
- **Duranton, Gilles.** 2014. "Agglomeration Effects in Colombia." *Working Paper, University of Pennsylvania, the Wharton School.*

- **Forero, Edgar.** 2003. "El Desplazamiento Interno Forzado en Colombia." *Conflict and Peace in Colombia: Consequences and perspectives for the Future.*
- **Friedberg, Rachel M.** 2001. "The Impact of Mass Migration on the Israeli Labor Market." *Quarterly Journal of Economics*, 1373–1408.
- **Ibáñez, Ana María.** 2008. *El Desplazamiento Forzoso en Colombia: Un Camino Sin Retorno Hacia la Pobreza.* Universidad de los Andes, CEDE, Ediciones Uniandes.
- **Ibáñez, Ana María, and Andrés Moya.** 2007. "La Población Desplazada En Colombia: Examen De Sus Condiciones Socioeconómicas Y Análisis De Las Políticas Actuales." *Informe final presentado a la Misión para el diseño de una estrategia para la reducción de la pobreza y la desigualdad (MERPD). Departamento Nacional de Planeación. Bogotá, Colombia.*
- **Ibáñez, Ana María, and Andrés Moya.** 2010. "Do Conflicts Create Poverty Traps? Asset Losses and Recovery For Displaced Households in Colombia." In *The Economics of Crime: Lessons for and from Latin America*. 137–172. University of Chicago Press.
- **Ibáñez, Ana María, and Carlos Eduardo Vélez.** 2008. "Civil Conflict and Forced Migration: The Micro Determinants and Welfare Losses of Displacement in Colombia." *World Development*, 36(4): 659–676.
- IDMC. 2015. Global Overview 2015. Internal Displacement Monitoring Center, Norwegian Refugee Council.
- **IPUMS.** 2014. *Integrated Public Use Microdata Series, International.* Minnesota Population Center, Minneapolis: University of Minnesota.
- Kondylis, Florence. 2010. "Conflict Displacement and Labor Market Outcomes in Post-war Bosnia and Herzegovina." *Journal of Development Economics*, 93(2): 235–248.
- Lafortune, Jeanne, José Tessada, and Ethan Lewis. 2014. "People and Machines: A Look at The Evolving Relationship Between Capital and Skill in Manufacturing 1850-1940 Using Immigration Shocks." *Working Paper*.
- Maystadt, Jean Francois, and Gilles Duranton. 2014. "The Development Push of Refugees: Evidence From Tanzania." *IFPRI Discussion Paper 01377*.
- Maystadt, Jean-François, and Philip Verwimp. 2014. "Winners and Losers among a Refugee-hosting Population." *Economic development and cultural change*, 62(4): 769–809.
- **Monras, Joan.** 2015. "Immigration and Wage Dynamics: Evidence From the Mexican Peso Crisis." *Discussion paper 2015-04, Sciences Po, Department of Economics.*
- Moretti, Enrico. 2011. "Local Labor Markets." Handbook of labor economics, 4: 1237–1313.
- Naranjo, Gloria. 2004. "Ciudades Y Desplazamiento Forzado En Colombia. El Reasentamiento De Hecho Y El Derecho Al Restablecimiento En Contextos Conflictivos De Urbanización." *Desplazamiento forzado. Dinámicas de guerra, exclusión y desarraigo,* 279–309.
- **Ottaviano, Gianmarco I.P., and Giovanni Peri.** 2012. "Rethinking the Effect of Immigration on Wages." *Journal of the European Economic Association*, 10(1): 152–197.

- **Peña, Liza Beatriz.** 2014. "Discrepancies in Labor Market Outcomes From Migration: Evidence From Colombia." *Working Paper, University of South Florida*.
- **Pérez Ortiz, Vólmar.** 2001. "El Desplazamiento Forzado por la Violencia en Colombia." In *Atención a los Desplazados: Experiencias Institucionales en Colombia.* 51–70. Pontificia Universidad Javeriana, Facultad de Ciencias Jurídicas, Departamento de Filosofía del Derecho, Bogotá.
- **Peri, Giovanni, and Vasil Yasenov.** 2015. "The Labor Market Effects of a Refugee Wave: Applying the Synthetic Control Method to the Mariel Boatlift." *NBER Working Paper No.* 21801.
- Ruiz, Isabel, and Carlos Vargas-Silva. 2013. "The Economics of Forced Migration." The Journal of Development Studies, 49(6): 772–784.
- Sánchez Torres, Fabio, and Ana María Díaz. 2005. "Los Efectos del Conflicto Armado en el Desarrollo Social Colombiano, 1990-2002." *Documento CEDE, Centro de Estudios sobre Desarrollo Económico, Facultad de Economía, Universidad de los Andes.*
- Silva Arias, Adriana Carolina, and Jaime Andrés Sarmiento Espinel. 2013. "Desplazados Forzados Y Su Participación En El Mercado Laboral Colombiano." *Revista Facultad de Ciencias Económicas*.
- Silva Arias, Adriana Carolina, and Juan Carlos Guataquí Roa. 2008. "Caracterización Instrumental Del Desplazamiento Forzado En Colombia: Hechos Estilizados Y La Encuesta Continua De Hogares (2001-2006)." *Revista de Ciencias Sociales*, 14(3).
- Smith, Christopher L. 2012. "The Impact of Low-skilled Immigration On The Youth Labor Market." *Journal of Labor Economics*, 30(1): 55–89.
- UNHCR. 2014. Global Trends 2013. United Nations High Comissioner for Refugees.

				·	<u> </u>	
	(1)		(2)	(2)		
	Full Sample		<secondary< td=""><td>Second</td><td>ary+</td></secondary<>		Second	ary+
	mean	sd	mean	sd	mean	sd
Male	0.63	0.48	0.67	0.47	0.54	0.50
Years of education	7.25	4.63	4.55	2.69	12.7	2.39
Age	36.9	12.9	38.4	13.7	34.1	10.5
Married (or civil union)	0.58	0.49	0.60	0.49	0.54	0.50
Urban	0.70	0.46	0.60	0.49	0.90	0.30
(Log of) Real wages	11.9	1.00	11.6	0.93	12.4	0.90
Observations	937964		535821		402143	

Table 1: Individual Summary Statistics (ECH Survey)

Notes: The table reports the weighted average of the individual characteristics in the ECH sample, for all years (1997-2004), using the survey weights. Column (2) summarizes variables for individuals with less than secondary school completed, and column (3) for those who have completed secondary school or above.

			<i>J</i>	(-)	- /		
	(1))	(2)		(3)	(4	.)
	Full Sat	ill Sample Lo		killed	High-Skilled		ID	Р
	mean	sd	mean	sd	mean	sd	mean	sd
Household characteristics:								
% Children in School	0.91	0.26	0.88	0.30	0.96	0.18	0.83	0.35
Electricity (0/1)	0.94	0.24	0.91	0.29	0.99	0.12	0.87	0.34
Water Supply (0/1)	0.83	0.38	0.75	0.43	0.94	0.24	0.74	0.44
Trash Collection $(0/1)$	0.75	0.43	0.64	0.48	0.93	0.25	0.70	0.46
Wealth index	-0.064	2.06	-0.79	1.93	1.06	1.71	-1.08	1.92
Household education (share adults):								
literate	0.93	0.20	0.89	0.24	1.00	0.058	0.90	0.23
with less than primary	0.069	0.20	0.11	0.24	0	0	0.093	0.22
with primary	0.35	0.37	0.57	0.31	0	0	0.46	0.39
with secondary	0.44	0.38	0.27	0.27	0.70	0.38	0.35	0.37
with tertiary	0.15	0.29	0.042	0.13	0.30	0.38	0.097	0.25
Household composition:								
% male	0.48	0.20	0.49	0.19	0.46	0.21	0.48	0.20
senior (65+)	0.14	0.42	0.15	0.41	0.14	0.42	0.100	0.35
adult (25-65)	2.03	0.87	2.16	0.93	1.84	0.72	1.90	0.81
young adult (15-24)	0.61	0.88	0.71	0.95	0.45	0.72	0.67	0.93
children (5-14)	1.72	0.92	1.84	1.00	1.53	0.74	1.94	1.08
babies (0-4)	0.54	0.78	0.62	0.86	0.41	0.64	0.71	0.88
HH age (average)	24.6	7.46	24.8	7.62	24.4	7.20	22.4	6.93
Observations	401155	<i>,</i> ,	287818	,	113337	,	6437	

Table 2: Household Summary Statistics (2005 Census)

Notes: The table reports the weighted average of household characteristics by skill group from the 2005 census. High skilled households are defined as those in which all adult members (of working ages 16-65) have secondary education or above. Columns 1-3 include only non-movers, defined as people who lived in the same municipality five years prior. Column (4) reports statistics for self-identified IDP households in the census.

indie 3: Municipality Summar	y Blatistics	
	(1)	
	All municipalities	
	mean	sd
IDP arriving between 2000-2004 (per 100 population)	4.78	7.84
Share urban	0.50	0.24
Population	76968.9	333095.2
1993 Census characteristics:		
Unclassified workers	0.23	0.097
Unemployed	0.034	0.027
Unpaid family worker	0.041	0.038
Self-employed	0.34	0.10
Male	0.51	0.021
Age	25.2	2.04
Migrants	0.35	0.17
Agricultural worker	0.14	0.073
Racial minority	0.039	0.15
Number of rooms	3.33	0.43
Dirt floor	0.24	0.20
School years	4.55	1.09
Kids in school	0.46	0.071
Wealth index	-0.67	0.85
Electricity	0.74	0.23
Water Supply	0.63	0.23
Trash collection	0.34	0.25
Population 1993	71901.5	276034.5
Additional controls:		
Population 1985	40958.3	102983.4
(Log of) Total area	6.53	1.14
Avg. Precipitation	3767.0	3210.9
Soil aptitude index	2.79	1.12
Erosion index	1.90	0.91
Latitude	-75.0	1.41
Longitude	5.79	2.62
Altitude	1072.1	856.9
Observations	518	

Table 3: Municipality Summary Statistics

Notes: The table reports the mean and standard deviation of variables used as controls at the municipal level, from the 2005 and 1993 census.





Figure 2: Number of IDP arrivals (2000-2004) per 100 population

		1	0 \	1 /		
	(1)	(2)	(3)	(4)	(5)	(6)
Enclave IV (t-1)	1.309***	1.331***	1.280***			
	(0.224)	(0.226)	(0.221)			
Enclave IV (1999-2003)				0.851***	0.839***	0.861***
				(0.158)	(0.161)	(0.149)
N	874574	490514	384060	208153	112489	95664
N-clusters	267	267	266	285	285	283
Skills	All	Low-skilled	High-skilled	All	Low-skilled	High-skilled
Specification	Short-run	Short-run	Short-run	Long-run	Long-run	Long-run
Municipal F.E.	Yes	Yes	Yes	No	No	No
Municipal Trends	Yes	Yes	Yes	No	No	No
Department F.E.	No	No	No	Yes	Yes	Yes
Municipal Controls	No	No	No	Yes	Yes	Yes

Table 4: IV First Stage (ECH Sample)

Notes: The table reports the first stage estimates from regressing the number of IDP arrivals on predicted arrivals using the enclave instrument. The short-run specification in columns 1-3 includes years 1997-2004. The long-run specification in columns 4-6 uses only 2004. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

Table 5: IV First Stage (2005 Census)								
	(1)	(2)	(3)					
Enclave IV (2000-2004)	0.654***	0.637***	0.672***					
	(0.134)	(0.139)	(0.125)					
N	851964	573077	278887					
N-clusters	517	517	517					
Skills	All	Low-Skilled	High-Skilled					
Specification	Long-run	Long-run	Long-run					
Department F.E.	Yes	Yes	Yes					
Municipal Controls	Yes	Yes	Yes					

Notes: The table reports the first stage estimates from regressing the number of IDP arrivals on predicted arrivals using the enclave instrument, using the 2005 Census. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

				0		
	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	IV	IV	IV
Share of displaced (t-1)	-0.00328	-0.00216	-0.00663*	-0.0142 [*]	-0.0151	-0.0115**
	(0.00404)	(0.00479)	(0.00376)	(0.00848)	(0.0103)	(0.00528)
N	874893	490798	384095	874574	490514	384060
N-clusters	268	268	267	267	267	266
F first stage				34.15	34.54	33.50
Skills	All	Low-skilled	High-skilled	All	Low-skilled	High-skilled
Specification	Short-run	Short-run	Short-run	Short-run	Short-run	Short-run
Municipal F.E.	Yes	Yes	Yes	Yes	Yes	Yes
Municipal Trends	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The table reports estimates from regressing the number of IDP arriving at t - 1 as a share of the total population, on the log of real wages for individuals in the ECH survey (1996-2004) at time t. All specifications include individual controls (gender, education, potential experience), municipal fixed effects, year fixed effects and municipal specific linear trends. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

|--|

	()			()	()		()	(0)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	OLS	OLS	OLS	OLS	ĪV	IV	IV	IV
Share of displaced (t-1)	0.000477	-0.00489	-0.00761	-0.00865	-0.0122	-0.00789	-0.0219**	-0.0170***
	(0.00532)	(0.00606)	(0.00529)	(0.00532)	(0.0108)	(0.0104)	(0.0109)	(0.00539)
N	303861	204238	186937	179857	303713	204214	186801	179846
N-clusters	268	266	268	266	267	265	267	265
F first stage					35.25	30.05	32.45	35.66
Skills	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled
Gender	Male	Male	Female	Female	Male	Male	Female	Female
Specification	Short-run	Short-run	Short-run	Short-run	Short-run	Short-run	Short-run	Short-run
Municipal F.E.	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipal Trends	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The table reports estimates from regressing the number of IDP arriving at t - 1 as a share of the total population, on the log of real wages for individuals in the ECH survey (1996-2004) at time t. All specifications include individual controls (education, potential experience), municipal fixed effects, year fixed effects and municipal specific linear trends. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

Table 8: Long-run effect on wages

	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	IV	IV	IV
Share of displaced (1999-2003)	0.00362*	0.00380*	0.00419**	0.00218	0.00188	0.00656
	(0.00190)	(0.00229)	(0.00202)	(0.00402)	(0.00446)	(0.00556)
N	208153	112489	95664	208153	112489	95664
N-clusters	285	285	283	285	285	283
F first stage				28.80	26.79	33.70
Skills	All	Low-skilled	High-skilled	All	Low-skilled	High-skilled
Specification	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run
Department F.E.	Yes	Yes	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The table reports estimates from regressing the number of IDP arriving between 1999-2003 as a share of the total population, on the log of real wages for individuals in the ECH survey in 2004. All specifications include individual controls (gender, education, potential experience), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

		0	Ľ	, , , , ,		,		
	(1) OLS	(2) OLS	(3) OLS	(4) OLS	(5) IV	(6) IV	(7) IV	(8) IV
Share of displaced (1999-2003)	0.00476**	0.00284	0.0000904	0.00600**	0.00643	0.00475	-0.0166**	0.00601
N	71002	50951	41487	44713	71002	50951	41487	44713
N-clusters	285	283	285	282	285	283	285	282
F first stage					27.52	32.84	25.76	34.92
Skills	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled
Gender	Male	Male	Female	Female	Male	Male	Female	Female
Specification	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run
Department F.E.	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table 9: Long-run effect on wages (by gender and skill)

Notes: The table reports estimates from regressing the number of IDP arriving between 1999-2003 as a share of the total population, on the log of real wages for individuals in the ECH survey in 2004. All specifications include individual controls (education, potential experience), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

Table 10: Long-run effect on wealth index								
	(1)	(2)	(3)	(4)	(5)	(6)		
	OLS	OLS	OLS	IV	IV	IV		
Share of displaced (2000-2004)	-0.00209	-0.00371**	0.00143	-0.0000947	-0.00103	-0.000474		
	(0.00161)	(0.00164)	(0.00204)	(0.00562)	(0.00588)	(0.00575)		
N	819362	552247	267115	819362	552247	267115		
N-clusters	517	517	517	517	517	517		
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled		
Specification	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run		
Department F.E.	Yes	Yes	Yes	Yes	Yes	Yes		
Municipal Controls	Yes	Yes	Yes	Yes	Yes	Yes		

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on the wealth index for households in the 2005 Census. All specifications include household head individual controls (gender, education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown p<0.10, p<0.05, p<0.01.

Table 11: Long-run	effect on w	vealth index ((by	gender a	and skills)
0			` '	0	,

	(1)	(2)	(3)	(4)
Share of displaced (2000-2004)	-0.000181	-0.000757	-0.00442	-0.000721
	(0.00651)	(0.00654)	(0.00532)	(0.00570)
Ν	420542	179953	131705	87162
N-clusters	517	517	517	517
F first stage	19.75	30.13	23.45	30.05
Skills	Low-Skilled	High-Skilled	Low-Skilled	High-Skilled
Household head gender	Male	Male	Female	Female
Department F.E.	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on the wealth index for households in the 2005 Census, by gender of the household head. All specifications include household head individual controls (education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	IV	IV	IV
Share of displaced (2000-2004)	0.000296	0.000231	0.000184	0.000803	0.000598	-0.000747
	(0.000339)	(0.000385)	(0.000318)	(0.00167)	(0.00151)	(0.00164)
N	851964	573077	278887	851964	573077	278887
N-clusters	517	517	517	517	517	517
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Specification	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run
Department F.E.	Yes	Yes	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes	Yes	Yes

Table 12: Long-run effect on access to electricity

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on access to electricity for households in the 2005 Census. All specifications include household head individual controls (gender, education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

Tuble 13. Long run cheer on access to electricity (by genuer and skins)									
	(1) (2) (3)								
	IV	IV	IV	IV					
Share of displaced (2000-2004)	0.000753	-0.00115	0.000796	-0.000216					
	(0.00162)	(0.00219)	(0.00150)	(0.00147)					
N	435198	187493	137879	91394					
N-clusters	517	517	517	517					
Skills	Low-Skilled	High-Skilled	Low-Skilled	High-Skilled					
Household head gender	Male	Male	Female	Female					
Department F.E.	Yes	Yes	Yes	Yes					
Municipal Controls	Yes	Yes	Yes	Yes					

Table 13: Long-run effect on access to electricity (by gender and skills)

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on access to electricity for households in the 2005 Census, by gender of the household head. All specifications include household head individual controls (education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

TT 11	т			• 1 1
Table 14.	Long-riin	ettect on	access to	niped water
10010 14.	Long run	chect on	access 10	piped mater

	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	IV	IV	IV
Share of displaced (2000-2004)	-0.00237**	-0.00234**	-0.00260*	-0.00295	-0.00332	-0.00759***
	(0.00103)	(0.000989)	(0.00138)	(0.00301)	(0.00263)	(0.00241)
Ν	851964	573077	278887	851964	573077	278887
N-clusters	517	517	517	517	517	517
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Specification	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run
Department F.E.	Yes	Yes	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on access to piped water for households in the 2005 Census. All specifications include household head individual controls (gender, education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

<u> </u>		1 1		/
	(1)	(2)	(3)	(4)
	IV	IV	IV	IV
Share of displaced (2000-2004)	-0.00255	-0.00627***	-0.00381	-0.00859***
	(0.00281)	(0.00240)	(0.00271)	(0.00249)
N	435198	187493	137879	91394
N-clusters	517	517	517	517
Skills	Low-Skilled	High-Skilled	Low-Skilled	High-Skilled
Household head gender	Male	Male	Female	Female
Department F.E.	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes

Table 15: Long-run effect on access to piped water (by gender and skills)

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on access to piped water for households in the 2005 Census, by gender of the household head. All specifications include household head individual controls (education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown p<0.10, p<0.05, p<0.05, p<0.01.

	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	IV	IV	IV
Share of displaced (2000-2004)	0.00108	0.000525	0.00246**	0.00836***	0.00602**	0.0105***
	(0.000704)	(0.000623)	(0.00104)	(0.00286)	(0.00296)	(0.00307)
N	851964	573077	278887	851964	573077	278887
N-clusters	517	517	517	517	517	517
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Specification	Long-run	Long-run	Long-run	Long-run	Long-run	Long-run
Department F.E.	Yes	Yes	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes	Yes	Yes

Table 16: Long-run effect on access to trash collection service

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on access to trash collection services for households in the 2005 Census. All specifications include household head individual controls (gender, education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

	(1)	(2)	(3)	(4)
	IV	IV	IV	IV
Share of displaced (2000-2004)	0.00599**	0.0117***	0.00707**	0.00900***
	(0.00273)	(0.00325)	(0.00284)	(0.00295)
Ν	435198	187493	137879	91394
N-clusters	517	517	517	517
Skills	Low-Skilled	High-Skilled	Low-Skilled	High-Skilled
Household head gender	Male	Male	Female	Female
Department F.E.	Yes	Yes	Yes	Yes
Municipal Controls	Yes	Yes	Yes	Yes

Table 17: Long-run effect on access to trash collection service (by gender and skills)

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on access to trash collection services for households in the 2005 Census, by gender of the household head. All specifications include household head individual controls (education, age), household controls (composition and education), department fixed effects, and municipal controls. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

Table 18: Impact on the rate of out-migration

	(1) OLS	(2) OLS	(3) OLS	(4) OLS	(5) IV	(6) IV	(7) IV	(8) IV
Share of displaced (t-1)	0.0327**	0.0512	0.0150	0.0553	0.282***	0.296*	0.226***	0.321*
	(0.0152)	(0.0443)	(0.0174)	(0.0447)	(0.0768)	(0.161)	(0.0814)	(0.180)
N	2660	2660	2660	2660	2645	2645	2645	2645
N-clusters	532	532	532	532	529	529	529	529
Skills	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled
Gender	Male	Male	Female	Female	Male	Male	Female	Female
F first stage					19.969	20.192	19.351	20.197

Notes: The table reports estimates from regressing the number of IDP arriving at time t - 1 as a share of the total population, on the rate of out-migration at time t. All specifications include municipality fixed effects, year fixed effects and the out-migration rate at time t - 1 as a control. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	OLS	OLS	OLS	OLS	ĪV	IV	ĪV	IV
Share of displaced (2000-2004)	0.0552	0.0359	0.0624*	0.115***	0.0880	0.165	0.140	0.297**
	(0.0350)	(0.0690)	(0.0348)	(0.0397)	(0.111)	(0.168)	(0.101)	(0.118)
N	516	516	516	516	516	516	516	516
Skills	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled	Low-skilled	High-skilled
Gender	Male	Male	Female	Female	Male	Male	Female	Female
F first stage					11.257	11.926	11.232	11.732

Table 19: Impact on the rate of out-migration: Long-run

Notes: The table reports estimates from regressing the number of IDP arriving between 2000-2004 as a share of the total population, on the rate of out-migration during the same period. All specifications include department fixed effects, municipal controls and the out-migration rate before 2000 (from birth) as a control. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown *p<0.10, **p<0.05, ***p<0.01.

		1	· ·	, ,,		
	(1)	(2)	(3)	(4)	(5)	(6)
Share of displaced (t-1)	0.0231	0.0186	-0.0101	-0.0163*	-0.0159	-0.0151**
	(0.0894)	(0.125)	(0.0678)	(0.00844)	(0.00972)	(0.00603)
N	605977	310949	295028	268597	179565	89032
N_clust	13	13	13	254	254	253
F_firststage	4.105	2.579	7.138	50.66	47.69	56.21
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Sample	Cities Only	Cities Only	Cities Only	No Cities	No Cities	No Cities

Table 20: Sample restrictions (by city)

Notes: The table reports the estimates from regressing the share of IDP arriving on the log of real wages from the DANE sample using the short-run specification. All specifications include municipality fixed effects, municipality specific trends, individual controls (gender, marital status, education, experience, rural). Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.10, ** p<0.05, ***p<0.01.

Table 21: Sample restrictions (by cell sample size)

	(1)	(2)	(3)	(4)	(5)	(6)
Share of displaced (t-1)	-0.0152	-0.0155	-0.0129**	-0.0297*	-0.0327	-0.0129
	(0.00938)	(0.0111)	(0.00574)	(0.0171)	(0.0222)	(0.00974)
N	786796	439763	347033	539092	299946	239146
N_clust	217	217	217	162	162	162
F_firststage	30.33	30.79	29.40	11.22	9.385	15.57
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Sample	Sample>0.05	Sample>0.05	Sample>0.05	Sample>0.1	Sample>0.1	Sample>0.1

Notes: The table reports the estimates from regressing the share of IDP arriving on the log of real wages from the DANE sample using the short-run specification. All specifications include municipality fixed effects, municipality specific trends, individual controls (gender, marital status, education, experience, rural). Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.10, **p<0.05, ***p<0.01.

		1		1 /		
	(1)	(2)	(3)	(4)	(5)	(6)
Share of displaced (t-1)	-0.0125	-0.0116	-0.0131**	-0.0363**	-0.0368*	-0.0317***
	(0.00807)	(0.00932)	(0.00577)	(0.0161)	(0.0208)	(0.0117)
N	868618	485900	382718	847500	469505	377995
N_clust	255	255	255	228	228	228
F_firststage	34.71	34.58	34.18	20.73	21.05	19.13
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Sample	Sample>50	Sample>50	Sample>50	Sample>100	Sample>100	Sample>100

Table 22: Sample restrictions (by cell sample size)

Notes: The table reports the estimates from regressing the share of IDP arriving on the log of real wages from the DANE sample using the short-run specification. All specifications include municipality fixed effects, municipality specific trends, individual controls (gender, marital status, education, experience, rural). Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.10, ** p<0.05, ***p<0.01.

		<i>y</i> 1	` ·	, , ,		
	(1)	(2)	(3)	(4)	(5)	(6)
Share of displaced (t-1)	-0.0506***	-0.0569**	-0.0343***	-0.0330**	-0.0372**	-0.0229**
	(0.0193)	(0.0239)	(0.0122)	(0.0138)	(0.0177)	(0.0105)
N	833786	463868	369918	790402	434735	355667
N_clust	265	265	264	264	264	263
F_firststage	20.17	20.31	19.55	26.08	25.98	26.06
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Sample	1999+	1999+	1999+	2000+	2000+	2000+

Table 23: Sample restrictions (by year)

Notes: The table reports the estimates from regressing the share of IDP arriving on the log of real wages from the DANE sample using the short-run specification. All specifications include municipality fixed effects, municipality specific trends, individual controls (gender, marital status, education, experience, rural). Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.10, ** p<0.05, ***p<0.01.

Table 24: Sample restrictions (by year)

		· 1	(, ,		
	(1)	(2)	(3)	(4)	(5)	(6)
Share of displaced (t-1)	-0.0207*	-0.0211	-0.0170**	-0.0137*	-0.0142	-0.00965**
	(0.0124)	(0.0138)	(0.00736)	(0.00768)	(0.00915)	(0.00448)
Ν	672571	383043	289528	484418	281894	202524
N_clust	264	264	263	207	207	207
F_firststage	31.52	35.83	24.84	47.55	53.06	34.12
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Sample	<2004	<2004	<2004	<2003	<2003	<2003

Notes: The table reports the estimates from regressing the share of IDP arriving on the log of real wages from the DANE sample using the short-run specification. All specifications include municipality fixed effects, municipality specific trends, individual controls (gender, marital status, education, experience, rural). Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.10, ** p<0.05, ***p<0.01.

			0 1			
	(1)	(2)	(3)	(4)	(5)	(6)
Share of displaced (t-1)	-0.0107*	-0.00910	-0.0108***	-0.0158	-0.0221 [*]	-0.0145
	(0.00647)	(0.00772)	(0.00417)	(0.0139)	(0.0131)	(0.0182)
N	305796	184861	120935	748643	341398	407245
N-clusters	189	189	189	264	261	264
F first stage	99.36	92.62	108.6	23.42	26.87	22.31
Skills	All	Low-Skilled	High-Skilled	All	Low-Skilled	High-Skilled
Years	98-01	98-01	98-01	01-04	01-04	01-04

Table 25: Year groups

Notes: The table reports the estimates from regressing the share of IDP arriving on the log of real wages from the DANE sample using the short-run specification. All specifications include municipality fixed effects, municipality specific trends, individual controls (gender, marital status, education, experience, rural). Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.10, ** p<0.05, ***p<0.01.

		Table 26:	Bounds			
	(1)	(2)	(3)	(4)	(5)	(6)
Share of displaced (t-1)	-0.00498	-0.00238	-0.00649	-0.0233**	-0.0196	-0.0360***
	(0.00748)	(0.00901)	(0.00549)	(0.0117)	(0.0124)	(0.00695)
N	867695	485094	382601	866317	489247	377070
N_clust	267	267	266	267	267	265
F_firststage	34.15	30.73	32.64	31.15	32.15	30.27
Skills	All	Low-Skilled	High-Skilled	All		
Bound	Low	Low	Low	High	High	High

Notes: The table reports the estimates from regressing the share of IDP arriving on the log of real wages from the DANE sample using the short-run specification. All specifications include municipality fixed effects, municipality specific trends, individual controls (gender, marital status, education, experience, rural). Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.10, ** p<0.05, ***p<0.01.

Tal	ble 27: Robı	ustness che	cks, impact	on piped wate	r access: Lo	w-skilled		
	ΞZ	(2) IV	(3) IV	(4) IV	(5) IV	(9) IV	(2) ا	(8) IV
Share of displaced (2000-2004)	-0.00308 (0.00262)	-0.00507* (0.00259)	0.00206 (0.00507)	-0.00391 (0.00273)	-0.00511* (0.00285)		-0.00404 (0.00315)	
Share of disp. (2000-2004) relative to 1993 pop						-0.00406 (0.00277)		
Net displacement								-0.00312 (0.00267)
Z	573077	282082	551865	553432	505159	573077	573077	573077
N-clusters	517	517	492	492	504	517	517	517
Excluded	None	Urban	Most arrivals	Most expulsions	13 cities			
Alternative						Pop share	Control expulsions	Net displacement
F first stage	20.869	23.344	17.141	21.073	19.029	13.613	12.890	20.195
Skills	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled
	-		-			-	-	

÷		
	_	
•	-	2
	V	
	U	
	- 1	
	ь,	
	<	2
	E	1
		1
	-	2
		I
	۰.	•
	u	1
	ē,	
	ų	ł
	α.	J
	~	1
	5	,
	٢.	l
	2	2
	π	1
	5	
	5	ï
	Ω.	ł
	1	1
	-	
	α	1
	5	
	1	
	-	1
	7	5
	5	1
	a	1
	-	1
	C	1
	-	1
•	-	
	C	١
	+	
	-	
	1	
	-	1
	-	
	+	
	С.	J
	-	
	π	1
	^	ŝ
	┶	1
	-	
	7	
	1	
	-	
		١
		•
	۲	1
	V	
	ř	1
	5	
	_	
	đ	j
	٩	
_		
-		
-	d U	
-		
-		
-	adu se	
-	ado ss	
-	ado sea	
-	Pess che	
-	ness che	
-	and seat	
-	stripess che	
-	stripes of the	
	istness che	
-	anstress che	
•	antstrices che	
-	plistness che	
	ohistness che	
	obitstness che	
	Anhiistness che	
- - -	Kobitstness che	
- - -	Anhistness che	
- -	· Kohistness che	
- - -	7. Kohiistness che	
- - -	T. Kohistness che	
- - f	or. Kohiisthess che	
- -	27. Kohiistness che	
- -	o JT. Kohiistness che	
- -	e 27. Kohiistness che	
- - -	le 27. Kohiistness che	
- - -	and 27. Kohiistness che	
- - -	hle 27. Kohiistness che	
	ahle 27. Kohiistness che	
	able 27. Kobiistness che	
- - - -	Jable 27. Kohiistness che	
- - -	Table 27. Kohiistness che	

The table reports estimates from regressing the number of IDP arriving in the previous five years as a share of the total population, on local access to piped water. All regressions use the preferred specification: include individual and household controls (education and demographic characteristics), municipal controls, geographic controls (altitude, latitude, arg. precipitation, soil erosion), and past outcome controls calculated from the 1993 census (education level, share agricultural workers, share of households with electricity, water supply, trash collection, kids in school, etc). See section 7.5 for details on robustness checks. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.05, ***p<0.05, ***p<0.01.

	(7) (8) IV IV	.00896*** 0.00282)		-0.00761*** (0.00244)	278887 278887	517 517		ol expulsions Net displacement	20.546 28.062	gh-skilled High-skilled	
h-skilled	(6) VI	φ.	-0.00628*** (0.00230)		278887	517		Pop share Contro	26.864	High-skilled Hig	tel sosulation on
r access: Hig	(5) IV	-0.00733*** (0.00241)			180541	504	13 cities		29.919	High-skilled	t off jo orego
n piped wate	(4) IV	-0.00665*** (0.00241)			273450	492	Most expulsions	I	31.196	High-skilled	the most of a
cks, impact o	(3) IV	-0.00906 (0.00611)			271389	492	Most arrivals		14.004	High-skilled	in the summer
ustness chec	(2) IV	-0.00704*** (0.00247)			235611	517	Urban		32.305	High-skilled	f IDD amining
able 28: Rob	(F)	-0.00749 ^{***} (0.00246)			278887	517	None		28.719	High-skilled	the minimum
Ţ		Share of displaced (2000-2004)	Share of disp. (2000-2004) relative to 1993 pop	Net displacement	Ν	N-clusters	Excluded	Alternative	F first stage	Skills	an increase many activities attaction of the pro-

Ę	9	ļ
-	2	
	ĩ) -
7	6	1
Ē	Ē	
1		
	ŝ	5
	G)
		5
	ł	
	4 F	
	3	
-	ç	5
	đ)
•	Ξ	1
	2	
	S	5
	t	,
	č	5
	ξ	
•	Ξ	
_	ŝ)
		j
-	č	
	с u	, ,
	ŭ	5
	Ĕ	1
	U	į
-	ç	2
,	ç	2
F		
¢	с Х	Ì
	٩)
-	c	2
F	π	1

The table reports estimates from regressing the number of IDP arriving in the previous five years as a share of the total population, on local access to piped water. All regressions use the preferred specification: include individual and household controls (education and demographic characteristics), municipal controls, geographic controls (altitude, longitude, latitude, arg. precipitation, soil erosion), and past outcome controls calculated from the 1993 census (education level, share agricultural workers, share of households with electricity, water supply, trash collection, kids in school, etc). See section 7.5 for details on robustness checks. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.05, ***p<0.05, ***p<0.01.

Tabl	le 29: Robus	stness check	ss, impact or	n trash collecti	ion access:	Low-skilled	H	
	(I)	(2) IV	(3) IV	(4) IV	(5) IV	(9) IV	(2) N	(8) IV
Share of displaced (2000-2004)	0.00614** (0.00272)	0.0102*** (0.00352)	0.0158*** (0.00556)	0.00523* (0.00268)	0.00617** (0.00309)		0.00644* (0.00333)	
Share of disp. (2000-2004) relative to 1993 pop						0.00763*** (0.00272)		
Net displacement								0.00624** (0.00276)
Z	573077	282082	551865	553432	505159	573077	573077	573077
N-clusters	517	517	492	492	504	517	517	517
Excluded	None	Urban	Most arrivals	Most expulsions	13 cities			
Alternative						Pop share	Control expulsions	Net displacement
F first stage	20.869	23.344	17.141	21.073	19.029	13.613	12.890	20.195
Skills	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled	Low-skilled
				;				

_ <u>e</u>
14
လု
Ś
2
୍ୟ
Π
i
ŝ
ല
_ଅ
ð
-
5
٠Ĕ
σ
ച
Ξ
2
5
g
H
-
ੁਸ਼ੂ
0
÷
ĕ
ã
Ē
.H
ා
Ť
ð
Ч
\circ
ŝ
- 83
Ē
÷
Ë
ق
0
\simeq
0
N
بە
5
صرً
Г

The table reports estimates from regressing the number of IDP arriving in the previous five years as a share of the total population, on local access to piped water. All regressions use the preferred specification: include individual and household controls (education and demographic characteristics), municipal controls, geographic controls (altitude, latitude, arg. precipitation, soil erosion), and past outcome controls calculated from the 1993 census (education level, share agricultural workers, share of households with electricity, water supply, trash collection, kids in school, etc). See section 7.5 for details on robustness checks. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.05, ***p<0.05, ***p<0.01.

	(8) IV			0.0106*** (0.00312)	278887	517		Net displacement	28.062	High-skilled	A minor trater
ı trash collection access: High-skilled	(2) IS	0.0109*** (0.00354)			278887	517		Control expulsions	20.546	High-skilled	24 330000 [000] 00 u
	(9) IV		0.0102 ^{***} (0.00251)		278887	517		Pop share	26.864	High-skilled	total nonulatio
	(5) IV	0.0109*** (0.00335)			180541	504	13 cities		29.919	High-skilled	there of the
	(4) IV	0.00942 ^{***} (0.00297)			273450	492	Most expulsions		31.196	High-skilled	, it man in the
s, impact on	(3) IV	0.0246*** (0.00826)			271389	492	Most arrivals		14.004	High-skilled	io the provie
stness checks	(2) IV	0.0101 ^{***} (0.00339)			235611	517	Urban		32:305	High-skilled	f IDD amining
le 30: Robus	ΈŊ	0.0105*** (0.00304)			278887	517	None		28.719	High-skilled	the number of
Tab		Share of displaced (2000-2004)	Share of disp. (2000-2004) relative to 1993 pop	Net displacement	Z	N-clusters	Excluded	Alternative	F first stage	Skills	a table wave actimates from wave

-P-
÷E
ť
Ł
6
Ξ
щ
ŝ
es
B
ă
Ц
<u>.</u> 9
- H
ğ
Ξ
5
Ч
S
H
7
<u>0</u>
Ļ,
20
ã
В
•=
က်
귕
ĕ
-9
Ś
ŝ
ŭ
st
Ę
9
ž
0
0
Ä
Ъ
H

The table reports estimates from regressing the number of IDP arriving in the previous five years as a share of the total population, on local access to piped water. All regressions use the preferred specification: include individual and household controls (education and demographic characteristics), municipal controls, geographic controls (altitude, longitude, latitude, avg. precipitation, soil erosion), and past outcome controls calculated from the 1993 census (education level, share agricultural workers, share of households with electricity, water supply, trash collection, kids in school, etc). See section 7.5 for details on robustness checks. Clustered standard errors at the municipal level are reported in parenthesis. Significance levels shown below *p<0.05, ***p<0.05, ***p<0.01.